Author's response to reviews

Title: Eating for 1, Healthy and Active for 2; feasibility of delivering novel, compact training for midwives to build knowledge and confidence in giving nutrition, physical activity and weight management advice during pregnancy.

Authors:

Andrea J Basu (partridgeandrea@hotmail.com)
Lynne Kennedy Prof (l.kennedy@chester.ac.uk)
Karen Tocque Prof (k.tocque@btinternet.com)
Sharn Jones (sharn.jones@wales.nhs.uk)

Version: 2
Date: 13 March 2014

Author's response to reviews:

Please see uploaded document addressing individual points raised by the reviewers.